Author's response to reviews

Title: Does computer use pose a hazard for future long-term sickness absence?

Authors:

Johan H Andersen (Johan.Hviid.Andersen@vest.rm.dk)
Sigurd Mikkelsen (simi@glo.regionh.dk)

Version: 3 Date: 16 March 2010

Author's response to reviews:

1. There are now 12 references, which all is cited in the text.
2. Table har beeb fit to A4.
3. Title page has been changed according to suggestion
4. Author list has been changes, and email address provided

Johan Hviid Andersen